Author's response to reviews

Title: Mucinous cystadenoma of the appendix mimicking cystic hydatid disease: a case report

Authors:

Andreas Krieg (andreas.krieg@med.uni.duesseldorf.de)
Jan Schulte am Esch 2nd (jan.schulteamesch@med.uni-duesseldorf.de)
Ludger W Poll (lpoll@gmx.de)
Stefan Braunstein (braunstein@med.uni-duesseldorf.de)
Wolfram T Knoefel (knoefel@med.uni-duesseldorf.de)

Version: 2 Date: 14 January 2008

Author's response to reviews: see over
Dear Prof. Kidd,

please find enclosed our manuscript " Mucinous cystadenoma of the appendix mimicking cystic hydatid disease: a case " for consideration for publication in Journal of Medical Case Reports.

We report of a 44-year old man with preoperative diagnosis of cystic hydatid disease. Surprisingly, explorative laparotomy revealed a mucinous cystadenoma of the appendix. We discuss the importance of reevaluation and differential diagnostic reflections of appendiceal mucocele.

We declare that the manuscript has not been published in another journal or is under consideration by any other journal or as part of a conference proceedings in a peer reviewed journal. We ensure that no material submitted as part of the manuscript infringes existing copyrights, or the rights of a third party. We warrant that all authors have contributed significantly to the work and have seen and approved the manuscript.

Please address all correspondence to me at the above named address

Yours sincerely,

Dr. med. Andreas Krieg